Author's response to reviews

Title: Piper betle induces phase I & II genes through Nrf2/ARE signaling pathway in mouse embryonic fibroblasts derived from wild type and Nrf2 knockout cells

Authors:

Wan Nuraini Wan Hasan (wnuraini_wh@yahoo.com)
Mi-kyoung Kwak (mkwak@catholic.ac.kr)
Suzana Makpol (suzana@medic.ukm.my)
Wan Zurinah Wan Ngah (zurina@medic.ukm.my)
Yasmin Anum Mohd Yusof (rahmatyasmin@yahoo.com)

Version: 2 Date: 29 July 2013

Author's response to reviews: see over
To

Tom Rowles PhD
Executive Editor
BMC Complementary and Alternative Medicine

Dear Editor

I am resubmitting the revised version of the paper:

MS: 1274448456910346

_Piper betle_ induces phase II genes and proteins through Nrf2/ARE signaling pathway in mouse embryonic fibroblasts cells in culture

Wan Nuraini Wan Hasan, Mi-kyoung Kwak, Suzana Makpol, Wan Zurinah Wan Ngah and Yasmin Anum Mohd Yusof

We have carefully revised the paper by answering all the questions raised by the 3 reviewers, and in some instances the answers are incorporated in the text where deemed appropriate (folder: point by point response).

We do hope that we are able to convince the reviewers that the work is now suitable for publication in BMC Complementary and Alternative Medicine.

We have asked a colleague who has been overseas for more than 10 years in seeking education and speaks English as a native English speaking person to help copyedit the paper.

Yours sincerely,

Yasmin Anum Mohd Yusof

Department of Biochemistry, Faculty of Medicine,
Universiti Kebangsaan Malaysia,
Jalan Raja Muda Abdul Aziz,
50300 Kuala Lumpur.